Author’s response to reviews

Title: HEALTH-RELATED QUALITY OF LIFE OF ADOLESCENTS WITH SICKLE CELL DISEASE IN SUB-SAHARAN AFRICA: A CROSS-SECTIONAL STUDY

Authors:

David Kambasu Muthahi (kambasumuthahi@gmail.com)
Joseph Rujumba (jrujumba@yahoo.com)
Hervé Monka Lekuya (helemenstar@yahoo.fr)
Deogratias Munube (deomunube@gmail.com)
Ezekiel Mupere (mupez@yahoo.com)

Version: 3 Date: 12 Apr 2019

Author’s response to reviews:

Editor Comments:

1. We note that the current submission contains some textual overlap with other previously published works, in particular:

   Background, from line 28 to 33


   Answer: Thank you. The correction has been done in Background section.

   Discussion, from line 14 to 23


   Answer: Thank you for your comments. Kindly find the correction done in the discussion section.

   Discussion, from line 54 (page 9) to line 6 (page 10)

   "Predictors of health-related quality of life in community-dwelling stroke survivors: a cohort study"
Answer: Thank you. We have re-phrasing this overlaps in discussion section.

Discussion, from line 28 to line 32


Answer: Thank you. The re-phrasing has been done in discussion section.

We are grateful for all your comments and orientations to make better this manuscript.

Thankyou very much.